Author's response to reviews

Title: Effect of an institutional development plan for user participation on professionals’ knowledge, practice, and attitudes. A controlled study.

Authors:

Marit By Rise (marit.b.rise@ntnu.no)
Hilde Grimstad (hilde.grimstad@ntnu.no)
Marit Solbjor (marit.solbjor@ntnu.no)
Aslak Steinsbekk (aslak.steinsbekk@ntnu.no)

Version: 4 Date: 1 November 2011

Author's response to reviews: see over
MS:1842982688540052. Formatting changes.
Effect of an institutional development plan for user participation on professionals' knowledge, practice, and attitudes. A controlled study.

We thank you for accepting our manuscript for publication.

The manuscript is now formatted according to all the remarks in the e-mail.

Yours sincerely

Marit By Rise /s/       Dr. Hilde Grimstad /s/

Dr. Marit Solbjor /s/   Dr. Aslak Steinsbekk /s/